Author's response to reviews

Title: Comparison of the medical students' perceived self-efficacy and the evaluation of the observers and patients

Authors:

Jette Ammentorp (Jette.Ammentorp@slb.regionsyddanmark.dk)
Janus L Thomsen (jlaust@health.sdu.dk)
Dorte E Jarbøl (DJarbol@health.sdu.dk)
René Holst (rholst@health.sdu.dk)
Anne L.H Øvrehus (aholm@health.sdu.dk)
Poul-Erik Kofoed (Poul.Erik.Kofoed@slb.regionsyddanmark.dk)

Version: 2 Date: 10 July 2012

Author's response to reviews:

Kolding, 10 July 2012

Dear Editor,

In the revised version of our paper we have now made the required changes to the manuscript; including an ethic statement and a section describing competing interested.

Hope that it is satisfactory so you will be able to proceed with peer review.

Sincerely yours,

Jette Ammentorp
Research Manager, Ph.D., MHSc